

Author's response to reviews

Title: Metformin Treatment in Diabetes and Heart Failure: When Academic Equipoise Meets Clinical Reality?

Authors:

Dean T Eurich (deurich@ualberta.ca)
Ross T Tsuyuki (ross.tsuyuki@ualberta.ca)
Sumit R Majumdar (me2.majumdar@ualberta.ca)
Finlay A Mcalister (finlay.mcalister@ualberta.ca)
Richard Lewanczuk (rlewancz@gpu.srv.ualberta.ca)
Marcelo C Shibata (Marcelo.shibata@ualberta.ca)
Jeffrey A Johnson (jeff.johnson@ualberta.ca)

Version: 5 Date: 5 February 2009

Author's response to reviews:

I have made the manuscript formatting as requested.

D.